Author's response to reviews

Title: Increasing children’s physical activity through a teaching-assistant led extracurricular intervention: Process evaluation of the Action 3:30 randomised feasibility trial

Authors:

Russell Jago (Russ.Jago@bristol.ac.uk)
Simon J Sebire (Simon.Sebire@bristol.ac.uk)
Ben Davies (Ben.Davies@bristol.ac.uk)
Lesley Wood (Lesley.Wood@bristol.ac.uk)
Kathryn Banfield (Kate.Banfield@bristol.ac.uk)
Mark J Edwards (M.J.Edwards@bristol.ac.uk)
Jane E Powell (jane.powell@uwe.ac.uk)
Alan A Montgomery (Alan.Montgomery@nottingham.ac.uk)
Janice L Thompson (j.thompson.1@bham.ac.uk)
Kenneth R Fox (K.R.Fox@bristol.ac.uk)

Version: 2
Date: 20 August 2014

Author's response to reviews:

The following formatting changes have been made:

1) We have added contact details for the lead author to the title page
2) We have added line numbering
3) Figures have been removed from the main document and titles removed from the figure files